Author's response to reviews

Title: Clinical decision support tools: analysis of online drug information databases

Authors:

Kevin A Clauson (clauson@nsu.nova.edu)
Wallace Marsh (wmash@su.edu)
Hyla Polen (hpolen@hcdpbc.org)
Matthew J Seamon (mseamon@nsu.nova.edu)
Blanca I Ortiz (blortiz@nsu.nova.edu)

Version: 5 Date: 6 March 2007

Author’s response to reviews: see over
Authors’ response to formatting requests

**Title:** Clinical decision support tools: analysis of online drug information databases (MS 1777467325120759)

**Authors:**
- Kevin A. Clauson: clauson@nsu.nova.edu
- Wallace Marsh: wmarsh@su.edu
- Hyla Polen: hpolen@hcdpbc.org
- Matthew J. Seamon: mseamon@nsu.nova.edu
- Blanca I. Ortiz: blortiz@nsu.nova.edu
Dr. Puebla,

We are submitting a revised copy of the manuscript, Clinical decision support tools: analysis of online drug information databases (MS 1777467325120759), in which we made modifications based on the requests about format. Specifically, qualifications were deleted, countries were added, and all vertical elements (lines and text) were removed from the Tables. No acknowledgements were added as there was nothing to include in that section.

Sincerely,

Kevin A. Clauson, Pharm.D.